ISSN 2392-1641 e-ISSN 2450-0097

# Economics and Business Review

Volume 10 (3) 2024

#### CONTENTS

Editorial introduction Monika Banaszewska, Michał Pilc

#### ARTICLES

Unveiling financial well-being: Insights from retired people in Third Age group in Poland, Spain and Denmark *Alicja Jajko-Siwek* 

Linder hypothesis and India's services trade Jadhav Chakradhar, Juhi Singh, Anusha Renukunta

The effect of output on employment in Poland during the COVID-19 pandemic Krzysztof Bartosik

CSR committees and their effect on green practices Ngoc Bao Vuong

The role of internationalisation in moderating the impact of ESG disclosure on financial performance Yuli Soesetio, Ely Siswanto, Subagyo, Muhammad Fuad, Dyah Arini Rudiningtyas, Siti Astutik

Price limit bands, risk-return trade-off and asymmetric volatility: Evidence from Tunisian Stock Exchange sectors Othman Mnari, Bassma Faouel

Quantile connectedness between social network sentiment and sustainability index volatility: Evidence from the Moroccan financial market Ahmed El Oubani

CEO pay ratio versus financial performance in Polish public companies Katarzyna Byrka-Kita, Karol Bulasiński

Innovation and Industry 4.0 in building the international competitiveness of food industry enterprises: The perspective of food industry representatives in Poland *Katarzyna Łukiewska* 

#### **Editorial Board**

Monika Banaszewska (Editor-in-Chief), Ivo Bischoff, Horst Brezinski, Gary L. Evans, Niels Hermes, Witold Jurek, Tadeusz Kowalski, Joanna Lizińska, Ida Musiałkowska, Paweł Niszczota, Michał Pilc (Deputy Editor-in-Chief), Konrad Sobański

#### International Editorial Advisory Board

Edward I. Altman – NYU Stern School of Business Udo Broll – School of International Studies (ZIS). Technische Universität. Dresden Conrad Ciccotello – University of Denver, Denver Wojciech Florkowski – University of Georgia, Griffin Oded Galor - Brown University, Providence Binam Ghimire – Northumbria University, Newcastle upon Tyne Christopher J. Green – Loughborough University Eduard Hochreiter - The Vienna Institute for International Economic Studies Mark J. Holmes - University of Waikato, Hamilton Andreas Irmen – University of Luxembourg Bruce E. Kaufman – Georgia State University, Atlanta Robert Lensink – University of Groningen Steve Letza – The European Centre for Corporate Governance Robert McMaster - University of Glasgow Victor Murinde – SOAS University of London Hugh Scullion – National University of Ireland, Galway Yochanan Shachmurove – The City College, City University of New York Thomas Taylor – School of Business and Accountancy, Wake Forest University, Winston-Salem Linda Goncalves Veiga – University of Minho, Braga Thomas D. Willett – Claremont Graduate University and Claremont McKenna College Habte G. Woldu - School of Management, The University of Texas at Dallas

#### Thematic Editors

Economics: Monika Banaszewska, Ivo Bischoff, Horst Brezinski, Niels Hermes, Witold Jurek, Tadeusz Kowalski, Ida Musiałkowska, Michał Pilc, Konrad Sobański • Finance: Monika Banaszewska, Gary Evans, Witold Jurek, Joanna Lizińska, Paweł Niszczota, Konrad Sobański • Statistics: Marcin Anholcer, Maciej Beręsewicz, Elżbieta Gołata

Language Editor: Owen Easteal, Robert Pagget

Paper based publication

© Copyright by Authors, Poznań 2024
 © Copyright for this edition by Poznań University of Economics and Business, Poznań 2024



This work is licensed under a Creative Commons Attribution 4.0 International License https://creativecommons.org/licenses/by/4.0

https://doi.org/10.18559/ebr.2024.3

ISSN 2392-1641 e-ISSN 2450-0097

POZNAŃ UNIVERSITY OF ECONOMICS AND BUSINESS PRESS ul. Powstańców Wielkopolskich 16, 61-895 Poznań, Poland phone +48 61 854 31 54, +48 61 854 31 55 https://wydawnictwo.ue.poznan.pl, e-mail: wydawnictwo@ue.poznan.pl postal address: al. Niepodległości 10, 61-875 Poznań, Poland

Printed and bound in Poland by: Poznań University of Economics and Business Print Shop

Circulation: 80 copies



# Economics and Business Review

Volume 10 (3) 2024

## CONTENTS

Editorial introduction	
Monika Banaszewska, Michał Pilc	3

#### ARTICLES

Unveiling financial well-being: Insights from retired people in Third Age group in Poland, Spain and Denmark Alicja Jajko-Siwek	7
Linder hypothesis and India's services trade Jadhav Chakradhar, Juhi Singh, Anusha Renukunta	34
The effect of output on employment in Poland during the COVID-19 pandemic Krzysztof Bartosik	58
CSR committees and their effect on green practices Ngoc Bao Vuong	84
The role of internationalisation in moderating the impact of ESG disclosure on financial performance Yuli Soesetio, Ely Siswanto, Subagyo, Muhammad Fuad, Dyah Arini Rudiningtyas, Siti Astutik	112
Price limit bands, risk-return trade-off and asymmetric volatility: Evidence from Tunisian Stock Exchange sectors Othman Mnari, Bassma Faouel	142
Quantile connectedness between social network sentiment and sustainability index volatility: Evidence from the Moroccan financial market Ahmed El Oubani	163
<b>CEO pay ratio versus financial performance in Polish public companies</b> Katarzyna Byrka-Kita, Karol Bulasiński	197
Innovation and Industry 4.0 in building the international competitiveness of food industry enterprises: The perspective of food industry representatives in Poland <i>Katarzyna Łukiewska</i>	216

# Quantile connectedness between social network sentiment and sustainability index volatility: Evidence from the Moroccan financial market<sup>1</sup>

# Ahmed El Oubani<sup>2</sup>

Abstract	Keywords
The goal of this paper is to investigate the connectedness between investor sentiment and volatility on the environ- mental, social, and governance index (ESG) in Morocco. Therefore, on the basis of an investor sentiment index con- structed from the X platform, and using quantile and fre- quency connectedness approaches, the findings reveal a sig- nificant connectedness between sentiment and ESG volatili- ty, particularly during turbulent events. Although sentiment acts most of the time as a net receiver of shocks, notably during the COVID-19 pandemic and geopolitical crises, it sometimes becomes a net sender of shocks. Furthermore, the spillovers between sentiment and ESG volatility are de- termined mainly by the long-term component, especially during extreme events, implying the persistence of shock transmission due to high uncertainty. The results also il- lustrate the impact of market conditions on the spillovers	• COVID-19 • ESG • frequency spillovers • quantile connectedness • sustainability

<sup>&</sup>lt;sup>1</sup> This article is a complementary version of the paper entitled 'Investor sentiment and sustainable investment: evidence from North African stock markets' accessible at: https://doi. org/10.1186/s43093-024-00349-x. The main contributions of this manuscript with respect to the previous manuscript are as follows. Firstly, while the previous manuscript develops an agregate sentiment index from three sentiment proxies, the present manuscript focuses only on the social media sentiment index. This allows us to compare the results from these sentiment indices used and to assess the robustness of social media sentiment as a representative indicator of investor sentiment and, consequently, its relevance in predicting investor behaviour and its impact on market risk. Secondly, while the previous manuscript integrates the frequency connectedness approach with the time connectedness approach, the present paper integrates the frequency connectedness approach with the practical implications of the findings for investors, companies and policy makers.

<sup>&</sup>lt;sup>2</sup> National School of Business and Management, Moulay-Ismail University, PO Box 3102 Toulal, Meknes, Morocco, a.eloubani@umi.ac.ma, https://orcid.org/0000-0002-0082-8671.

between sentiment and ESG volatility. The conclusions of this study provide useful guidance for pro-ESG investors, policymakers, and companies.

JEL codes: G11, G15, G41, N27, N57.

Article received 5 March 2024, accepted 8 August 2024.

**Suggested citation:** El Oubani, A. (2024). Quantile connectedness between social network sentiment and sustainability index volatility: Evidence from the Moroccan financial market. *Economics and Business Review*, *10*(3), 163–196. https://doi.org/10.18559/ebr.2024.3.1200



This work is licensed under a Creative Commons Attribution 4.0 International License https://creativecommons.org/licenses/by/4.0

# Introduction

The growing awareness of climate change and sustainability issues has prompted many stock markets to introduce sustainability indices. These indices cover only companies that comply with specific environmental, social, and governance (ESG) criteria. Recently, many investments have been reallocated to ESG assets (Gao et al., 2022; Zhan & Santos-Paulino, 2021), representing a high proportion of global equity portfolios (Daugaard, 2020). Therefore, investors' increasing attention to ESG-related stocks might have a positive effect on companies' financial performance (Vuong, 2022; Zhang & Zhang, 2023) and on the returns of ESG indices on financial markets (El Ouadghiri et al., 2021; López-Cabarcos et al., 2019).

Although ESG indices are derived from general indices, they might be more sensitive to market fluctuations than general indices are (Ortas et al., 2014). This is explained by investors' sensitivity to sustainable investment. Some studies have analysed the relationship between sentiment and sustainability indices on the basis of different investor sentiment proxies (El Ouadghiri et al., 2021; Giannarakis et al., 2016; López-Cabarcos et al., 2019; Pitoska et al., 2017). Giannarakis et al. (2016) and Pitoska et al. (2017) revealed that consumer sentiment has a positive effect on sustainability indices. On the basis of the Google search volume index, El Ouadghiri et al. (2021) reported that investors' attention to climate change and pollution has a significant and positive effect on sustainability index returns. López-Cabarcos et al. (2019) indicated that social network sentiment has a stronger effect on the S&P 500 Environmental & Socially Responsible Index than on the S&P 500 Index. Using

event study analysis, Nyakurukwa and Seetharam (2023a) found that investors react strongly to positive ESG news, whereas there is no statistically significant reaction to negative ESG news sentiment.

Investors' overreaction to ESG news could have a significant effect on ESG volatility and risk. However, the relationship between ESG performance and the volatility of ESG returns remains unclear: some studies have shown that ESG performance reduces risk (Li et al., 2022; Shakil, 2021), whereas others have revealed that it increases risk (Khan et al., 2024; Nofsinger et al., 2019). Thus, examining the connectedness between investor sentiment and ESG index volatility can provide a better understanding of this relationship. Furthermore, limited effort has been devoted to investigating the quantile, time, and frequency connectedness between sustainability sentiment and ESG index volatility. Moreover, despite the growing interest in ESG stock indices, little research exists on the connectivity between ESG sentiment and volatility in emerging markets. On the basis of a systematic literature review, Daugaard (2020) noted that there is very little literature on ESG investment in emerging markets. In addition, no research has addressed this connectivity in the Moroccan context, where it is unclear whether the results of previous research apply.

To fill these gaps, this article aims to investigate the dynamic connectedness between the X-based sustainable investment sentiment index and the volatility of the ESG index in the Moroccan financial market. Using quantile and frequency connectedness approaches, the main empirical results show that the connectedness between investor sentiment and ESG index volatility varies over time and depends on certain events. Furthermore, the results reveal that sentiment is a net receiver of shocks from ESG index volatility most of the time, but sometimes becomes a net transmitter of shocks, implying a feedback effect between sentiment and ESG index volatility. The net receipt of shocks is more pronounced during periods of extreme events. Moreover, most of the time long-term spillovers dominate short-term spillovers, especially during distress events such as COVID-19 and geopolitical tensions. With respect to market conditions, the findings show that spillover effects are greater in the upper quantiles than in the other quantiles, and that sentiment is a net transmitter of shocks to ESG index volatility in the upper guantiles, particularly in times of crisis.

We make four main contributions to the literature. Firstly, most research examining ESG investment issues has focused primarily on the US and Europe, where the ESG concept first emerged (Demers et al., 2021; Engelhardt et al., 2021). Nevertheless, related research on certain emerging markets is in noticeably short supply, including countries that are highly vulnerable to climate change and where ESG investment is an extremely critical issue. Morocco is an emerging country that is highly vulnerable to climate change, and where authorities are striving to encourage ESG investments and achieve sustainability goals. Against this backdrop, the Casablanca Stock Exchange recently introduced the ESG index. It is therefore relevant to examine the relationship between investor behaviour and ESG investment in Morocco to determine whether the findings of previous studies apply to Morocco, where the ESG concept has gained momentum in recent years. This study is the first to explore the dynamic and complex interplay between social media sentiment and ESG index volatility in the Moroccan ESG market. We thereby examine whether investors in emerging markets, particularly in Morocco, are committed to sustainable investment and how their behaviour may affect the ESG market. Moreover, this study provides a better understanding of the progress of sustainable markets on the African continent, as the Moroccan financial market is one of the largest in Africa and attracts foreign investors, which can offer investors opportunities for geographic diversification. Secondly, to the best of our knowledge, this is also the first study to examine the bidirectional effect between social media sentiment and volatility on the ESG index, providing a comprehensive examination of this relationship. Thirdly, the sample in this study includes the most recent major events, namely, the COVID-19 epidemic, and the most recent major geopolitical events, such as the Russian-Ukrainian and Israeli-Palestinian conflicts, allowing us to better understand how these turbulent events influence the relationship between sentiment and the sustainability index, and thus help manage the risk propagated by these events. Fourthly, this study extends the literature on the relationship between ESG sentiment and volatility by using a fairly novel connectedness method, the dynamic QVAR model proposed by Ando et al. (2022), and by including frequency connectedness in our analysis. This enables us to capture connectivity in the frequency and time domains, as well as the quantile connectivity representing different market states. This is founded on the idea that the extent of connectivity can vary depending on whether markets are bearish, bullish or normal, depending on the time horizon (short or long term), and can also vary over time owing to extreme events. As a result, this allows us to understand the impact of the interaction between heterogeneous investor behaviour (short-term versus long-term investors) on the ESG market, as well as the ESG market risk at different quantiles (bearish, bullish and normal market conditions), which is more beneficial for the management of portfolios and the supervision of financial systems (Liao & Pan, 2022; Londono, 2019). This is especially important as financial markets exhibit structural breaks in return series (Cunado et al., 2023; Suleman et al., 2024) and nonlinearity caused by investor psychology (Lekhal & El Oubani, 2020).

The remainder of this paper is organised as follows. Section 1 reviews recent related literature. Section 2 outlines the methodology and data. Section 3 presents and analyses the empirical results of the paper. Section 4 performs robustness tests. Section 5 discusses the implications of the study. The last section presents conclusions.

# 1. Literature review and hypothesis development

With a market capitalisation of \$35.30 trillion in 2020, ESG investments are expected to reach \$50 trillion by 2025 (Lei et al., 2023). This reflects investors' preference for socially responsible investments that focus more on social or ethical utility (Gao et al., 2022; Garel & Petit-Romec, 2021), especially as such investments might be more resilient to turbulent events. Indeed, sustainable investment might continue to increase even in times of crisis, suggesting that investors and companies can be rewarded during crises for environmental sustainability (Garel & Petit-Romec, 2021). This can be explained by the fact that the ESG strategy might strengthen the brand image and loyalty of responsible companies (Pedini & Severini, 2022), helping attract loyal managers, investors, and customers (Boubaker et al., 2022; Isaak & Lentz, 2020). Loyal managers possess desirable capabilities for managing businesses, employees and customer relationships, which could facilitate improved productivity and profitability (Pedini & Severini, 2022). Loyal investors are motivated by nonpecuniary incentives to invest in ESG assets and are therefore unlikely to sell their investments, even in times of crisis (Hartzmark & Sussman, 2019). Loyal customers offer socially responsible companies financial support to overcome the crisis, resulting in higher cumulative abnormal returns (Boubaker et al., 2022). Moreover, firms that report ESG information benefit from financial support (Raimo et al., 2021). In fact, rating agencies have adhered to the principle of responsible investment, where ESG issues are included in their rating methodologies, implying that companies with higher ESG performance receive higher ratings (Bannier et al., 2022) and therefore benefit from lower interest rates. For instance, Phillips has received a EUR 1 billion loan from a syndicate of banks on the basis of sustainability (Klink & Gonciarenko, 2017). As a result, ESG performance could improve companies' resistance to risks. In this vein, Wang et al. (2023) argued that increasing environmental uncertainty drives companies to undergo a green transformation, improving their resistance to risks while enhancing their ESG performance. Li et al. (2022) revealed that ESG ratings reduce a firm's default risk. Murata and Hamori (2021) reported that ESG disclosures lower the risk of future stock price crashes. Thus, the high value placed on ESG assets is explained by their ability to hedge against downside risk during periods of market volatility (Albuquerque et al., 2020; Broadstock et al., 2021).

On the other hand, investor psychology might cause mispricing and high market volatility, which calls into question the efficiency market hypothesis. According to this hypothesis, prices instantly incorporate all available information, because investors can process all this information rationally (Fama, 1970). Nonetheless, investors' rationality is limited (Simon, 1982), and investors are affected by psychological biases that can influence their decision-making (El Oubani, 2023; Tversky & Kahneman, 1974; Willett, 2024). Behavioural finance studies have shown that investor sentiment can affect stock prices and explain many market anomalies, such as excessive volatility (Chiu et al., 2018; El Oubani & Lekhal, 2022; Lekhal & El Oubani, 2020; Paramanik & Singhal, 2020; PH & Rishad, 2020; Shiller, 1981; Wang et al., 2021).

Consequently, investor sentiment has a major impact on financial markets, and ESG markets might not be an exception or immune to this impact. In fact, investor sentiment could influence the ESG market, leading to market inefficiency and thus price deviation from the intrinsic value. Khan et al. (2024) reported that improvements in ESG profiles increase market prices relative to their true value and that market sentiment plays a moderating role in the link between ESG performance and mispricing, suggesting that ESG performance is a frictional factor for market efficiency. Bofinger et al. (2022) noted that socially responsible stocks tend to be overvalued. Liu et al. (2022) reported that ESG sentiment is positively associated with the volatility risk premium, particularly the impact of environmental and social factors.

The challenge in these studies lies in the fact that investor sentiment is unobservable and must thus be estimated from proxies. These proxies can be divided into three categories, namely, market-based measures, survey-based measures, and text-based measures. Market and survey-based measures are the most traditional methods, but their weaknesses arise from the fact that they indirectly attribute investor sentiment to a particular asset and are measured at low frequencies, which might not explain stock prices determined at high frequencies. The textual method is a relatively new way of extracting sentiment from news texts or social media posts. Accordingly, a large body of research uses social media and news-based sentiment in connection with ESG returns, trading volume and volatility. Through bibliometric analysis, Nyakurukwa and Seetharam (2023b) revealed that social media has emerged as an important facet of financial market understanding. Several measures of social media sentiment have been used in this context, including sentiments derived from the X platform (Abdollahi et al., 2024; Shutes et al., 2016), Facebook (Siganos et al., 2017) and Reddit (AlZaabi, 2021). Many studies confirm the impact of social media sentiment on returns and volatility. Indeed, Nyakurukwa and Seetharam (2024) reported a more powerful effect of social media sentiment on stock market returns in the US than did news media. Using textual analysis from news and social media, Abdollahi et al. (2024) found that media sentiment induces market volatility. Haroon and Rizvi (2020) showed that media-generated panic is correlated with increased volatility in global stock markets. This evidence suggests that part of financial market volatility is connected to media sentiment. As a result, the following hypothesis is formulated:

**Hypothesis 1.** There is a significant connectedness between social media sentiment and ESG volatility.

Within this connection, ESG volatility can transmit shocks to investor sentiment, which in turn can transmit shocks to ESG volatility. Dhasmana et al. (2023) asserted that the ESG index influences investor sentiment. In fact, as the ESG index increases, investor sentiment weakens; in contrast, poor ESG index performance strengthens investor sentiment. Ford et al. (2022) revealed that optimistic investor sentiment is driven by the highest-rated ESG portfolio. Chen and Yang (2020) found that investors are optimistic (pessimistic) about companies with higher (lower) ESG scores. Therefore, ESG performance or risk could influence investor sentiment.

Reciprocally, investor sentiment can play a dominant role in the ESG market. Indeed, Sabbaghi (2022) observed that the volatility of ESG companies is greater for bad news than for good news. Yu et al. (2023) reported a significant negative relationship between ESG news sentiment and stock market crash risk, suggesting that higher sentiment of ESG news could mitigate the risk of stock market crashes. Specifically, ESG sentiment impacts stock market crash risk by lowering negative ESG incidents, information asymmetry and agency costs. In addition, Liu et al. (2022) noted that news-based ESG sentiment influences the equity volatility risk premium. Wan et al. (2024) noted that high attention to ESG results in high volatility because greater attention to ESG information can lead to a diversity of opinions and disagreements between investors in ESG investment, thus inducing an increase in volatility connectedness. Cheng et al. (2024) pointed out that companies with better-quality sustainability reports receive greater coverage from analysts and positive reactions from investors on stock prices, which is likely to lead to greater volatility. Abdollahi et al. (2024) observed a bidirectional connection between social media sentiment and volatility.

On the basis of the aforementioned literature, we propose the following hypothesis:

**Hypothesis 2.** There is a feedback channel between investor sentiment and ESG index volatility.

The link between sentiment and ESG volatility can be more pronounced in times of crisis. Indeed, in times of crisis, investors prefer to switch from conventional to sustainable investments, which can cause large movements in stock markets (Wan et al., 2024). Consequently, ESG investments might not hedge against the market shocks propagated by distress events, such as the COVID-19 pandemic and geopolitical risk. Nofsinger and Varma (2014) demonstrated that risk contagion is likely to be more severe during periods of turbulence, including in the ESG market. Döttling and Kim (2022) reported a high sensitivity of sustainable investments to return shocks during crises, especially the

COVID-19 outbreak. In fact, exposure to COVID-19 positively and significantly impacts stock market volatility (Liu et al., 2023). This can be explained by investor perceptions and sentiment, as greater exposure to COVID-19 can be seen as a risk factor, resulting in greater selling pressure and volatility, especially as social media coverage of the COVID-19 outbreak (Umar et al., 2021) increased the sentiment of fear. In this context, Chen and Lin (2022) argued that ESG investment does not hedge against the risks that financial markets commonly experience, as their performance and volatility can be influenced by a variety of extreme events. This contradicts the findings of Albuquerque et al. (2020), who documented that stocks which are better rated in terms of ESG performance showed significantly lower volatility during the spike of the COVID-19 epidemic, because of investor and customer loyalty to sustainability. This also disagrees with Rubbaniy et al. (2022), who concluded that during the pandemic, investors might have shifted their portfolios from risky assets to ESG assets considered to be safe assets.

In addition to COVID-19, geopolitical risks can also have a significant effect on sustainable investment. These risks include the potential disturbance of normal international relations caused by several factors, such as shocks and tensions between nations (Ren et al., 2023), military conflicts, and war and terrorism threats (Caldara & Iacoviello, 2022). Yang et al. (2024) reported that changes in the geopolitical risk index bring about changes in ESG stocks. Sohag et al. (2022) highlighted the negative impact of geopolitical risk shocks on green equity. In this context, investor sentiment can explain the relationship between geopolitical risk and volatility. Guo and Shi (2024) examined the impact of investor sentiment and China-US geopolitical risk on Chinese stock market volatility and found that the interplay effects of China-US geopolitical risk and investor sentiment can further influence industry stock market volatility.

Building on this foundation, we posit the following:

**Hypothesis 3.** Disruptive events can amplify the connectedness between sentiment and ESG index volatility, creating increased risk in the sustainable market.

Volatility tends to be highly persistent (Wei et al., 2022), a property that is considered a stylised fact of financial markets known as long memory (Shi & Ho, 2015). Investor sentiment could be responsible for this volatility persistence. In fact, the uncertainty that prevails in distressed situations can transmit shocks from investor sentiment to market volatility over longer periods, which can result in low-frequency connectivity. Wu et al. (2023) showed that low-frequency connectedness tends to dominate during conflicts. Abdollahi et al. (2024) found that social media sentiment has a stronger lasting effect than news sentiment for some markets. Wan et al. (2024) found that high attention primarily impacts volatility connectedness at low frequencies. These results are in line with those found in the ESG market. Indeed, Gao et al. (2022) found that medium-frequency volatility connectivity between ESG stock indices from eight regions dominates. This might be explained by the fact that ESG stock indices generate less sudden short-term risk and take longer to absorb risk in the risk transmission process (Wan et al., 2024). El Oubani (2024) found significant connectedness between sentiment and ESG market volatility, and this connectedness is due mainly to the long-term component, especially during extreme events.

On the basis of this discussion, the following hypothesis is formulated:

**Hypothesis 4.** The connectedness between sentiment and ESG index volatility is created at low frequencies, particularly during extreme events leading to high persistent volatility.

The connectedness between ESG sentiment and volatility might depend on market conditions, which can be captured by calculating this connectedness at different quantiles. In this respect, using the quantile connectedness approach to examine the relationship between sentiment and ESG volatility across different quantiles can be highly relevant. Indeed, conditional mean connectedness may provide limited information on the connectedness between sentiment and ESG volatility as financial markets can experience structural breaks (Cunado et al., 2023). Specifically, the most essential information on negative and positive shocks in financial time series is available in the lower and upper extreme quantiles (Londono, 2019). On the basis of quantile connectedness estimates, Sheikh et al. (2024) showed that the overall connectedness between daily trade policy uncertainty, geopolitical risk, the global financial stress index and the three Australian financial markets (i.e. conventional, sustainable and Islamic stock markets) increases at extreme bearish and bullish quantiles relative to normal market conditions. Guo and Shi (2024) reported that the impact of the interaction effects of Chinese and US geopolitical risk and investor sentiment on market volatility is highly heterogeneous, and that its magnitude lies mainly in the upper and lower tails. El Khoury et al. (2024) found that dynamic spillover effects are greater under extreme market conditions than normal conditions. With respect to the ESG market, Yang et al. (2024) noted that the predictability of ESG returns by geopolitical risk is asymmetric across different ESG return distributions. Dhasmana et al. (2023) found that the relationship between investor sentiment and the ESG index is asymmetric and influenced by extreme market conditions. Therefore, studying the impact of different market conditions can provide in-depth information on the link between sentiment and ESG index volatility, and provide investors with valuable information that could improve their decision-making by trading according to different market conditions, including upper and lower quantiles.

On the basis of the above literature review, we posit the following:

**Hypothesis 5.** The magnitude of the spillover effects between investor sentiment and ESG index volatility depends on market conditions.

# 2. Methodology and data

### 2.1. Methodology

Some approaches consider the mean-based VAR (Diebold & Yilmaz, 2012; Baruník & Křehlík, 2018) to examine connectedness between variables, but these approaches do not allow us to assess whether co-movement between variables in the network depends on the extreme quantiles and whether the shock is positive or negative. Furthermore, as the most essential information on negative and positive shocks in financial time series is available at the lower and upper extremes (Londono, 2019), these approaches do not provide sufficient information on the connectedness between sentiment and ESG volatility, especially as financial markets can experience structural breaks (Cunado et al., 2023). Accordingly, in our study we use the quantile connectedness approach developed by Ando et al. (2022), which is an improvement over the mean-based measures version. Quantile connectedness considers both extreme positive structural shocks (upper quantiles) and extreme negative structural shocks (lower quantiles). This allows us to see how normal (median), bullish (right-tail) and bearish (left-tail) market conditions affect spillover effects between variables.

To compute the connectedness measures at each quantile  $\tau$ , we first estimate a quantile vector autoregression, QVAR(p), which is as follows:

$$y_{t} = \mu(\tau) + \sum_{j=1}^{p} \Phi_{j}(\tau) y_{t-j} + u_{t}(\tau)$$
(1)

in which  $y_t$  and  $y_{t-j}$ , j = 1, ..., p are  $N \times 1$  dimensional endogenous variable vectors,  $\tau$  denotes the quantile of interest and lies within [0,1], p is the lag length of the QVAR model,  $\mu(\tau)$  stands for the  $N \times 1$  dimensional conditional mean vector,  $\sum_{j=1}^{p} \Phi_j(\tau)$  is an  $N \times N$  dimensional QVAR coefficient matrix, and  $u_t(\tau)$  represents the  $N \times 1$  dimensional error vector that has an  $N \times N$  dimensional error variance—covariance matrix  $\sum(\tau)$ . To transform the QVAR(p) into its quantile vector moving average representation,  $QVMA(\infty)$ , we employ Wold's theorem:

$$y_{t} = \mu(\tau) + \sum_{j=1}^{p} \Phi_{j}(\tau) y_{t-j} + u_{t}(\tau) = \mu(\tau) + \sum_{i=0}^{\infty} \Psi_{i}(\tau) u_{t-i}$$
(2)

Then, we calculate the generalised forecast error variance decomposition (GFEVD) at the forecast horizon H (Koop et al., 1996; Pesaran & Shin, 1998).

The GFEVD indicates the impact of a shock in variable j on variable i with respect to its forecast error variance share. The GFEVD is given by the following formula:

$$\theta_{ij}^{g}(H) = \frac{\sum_{h=0}^{M-1} \left( e_{i}^{'} \Psi_{h}(\tau) \sum_{j} (\tau) e_{j} \right)^{2}}{\sum_{h=0}^{M-1} \left( e_{i}^{'} \Psi_{h}(\tau) \sum_{j} (\tau) \Psi_{h}(\tau)^{'} e_{j} \right)^{2}}$$
(3)

in which  $e_i$  represents a zero vector with unity at the  $i^{th}$  position. In the decomposition matrix, the normalisation of elements is as follows:

$$\tilde{\theta}_{ij}^{g}(H) = \frac{\theta_{ij}^{g}(H)}{\sum_{j=1}^{k} \theta_{ij}^{g}(H)}$$
(4)

The normalisation is expressed by the following equations:

$$\sum_{j=1}^{K} \tilde{\theta}_{ij}^{g}(H) = 1 \quad \text{and} \quad \sum_{i,j=1}^{K} \tilde{\theta}_{ij}^{g}(H) = K$$
(5)

Subsequently, according to Diebold and Yilmaz (2012), all GFEVD-based connectedness metrics are estimated as follows. The total connectedness index (*TCI*) provides the average level of total spillover and is written as follows:

$$TCI = K^{-1} \sum_{i,j=1, i \neq j}^{K} \tilde{\theta}_{ij}^{g}(H)$$
(6)

This measure can be seen as an indicator of market uncertainty. The total directional connectedness TO is defined as:

$$TO_{i} = \sum_{j=1, i \neq j}^{K} \tilde{\theta}_{ji}^{g}(H)$$
(7)

The total directional connectedness *FROM* is as follows:

$$FROM_{i} = \sum_{j=1, i \neq j}^{K} \tilde{\theta}_{ij}^{g}(H)$$
(8)

The NET total directional spillover is written in the following form:

$$NET_i(H) = TO_i - FROM_i$$
(9)

If  $NET_i > 0$ , this indicates that variable *i* influences all the other variables more than it is influenced by them. The variable *i* is therefore a net transmitter of shocks. It is a net receiver of shocks when  $NET_i < 0$ .

The net pairwise connectedness (NPDC) is computed as follows:

$$NPDC_{ii}(H) = \tilde{\theta}^{g}_{ii}(H) - \tilde{\theta}^{g}_{ii}(H)$$
(10)

If  $NPDC_{ij}(H) > 0$ , variable *j* has a greater effect on variable *i* than variable *i* has on variable *j*. Consequently, the variable *j* dominates the variable *i*. If  $NPDC_{ii}(H) < 0$ , the variable *i* dominates the variable *j*.

We integrate the frequency domain connectedness approach of Baruník and Křehlík (2018) with Ando et al.'s *QVAR* approach (2022). The frequency domain separates spillovers into high-frequency (short-term) and low-frequency (long-term) spillovers. The former indicates that connectedness results from shocks that have a short-term effect on the system, whereas the latter suggests that connectedness results from shocks that have a long-term effect on the variables in the system.

According to Baruník and Křehlík (2018),  $\Psi(e^{-i\omega}) = \sum_{h} e^{-i\omega h} \Psi_{h}$  is the frequency response function generated from the Fourier transform of the coefficient  $\Psi_{h}$  with  $i = \sqrt{-1}$ . The generalised causation spectrum over frequencies  $\omega \in (-\pi, \pi)$  is determined as:

$$\theta_{ij}(\omega) = \frac{\sigma_{jj}^{-1} \left| \Psi(e^{-i\omega}) \sum_{ij} \right|^2}{\left( \Psi(e^{-i\omega}) \sum \Psi'(e^{+i\omega}) \right)_{ii}}$$
(11)

in which  $\Psi(e^{-i\omega}) = \sum_{h} e^{-i\omega h} \Psi_{h}$  is the Fourier transform of the impulse response  $\Psi_{h'}$  and  $\theta_{ij}(\omega)$  denotes the fraction of the  $i^{th}$  variable at a specified frequency  $\omega$  resulting from shocks to the  $j^{th}$  variable.

The generalised variance decomposition at a given frequency band d = (a, b) is as follows:

$$\theta_{ij}(d) = \frac{1}{2\pi} \int_{a}^{b} \Gamma_{i}(\omega) \,\theta_{ij}(\omega) \,d\omega \tag{12}$$

where  $\Gamma_i(\omega)$  denotes the power of the *i*<sup>th</sup> variable at a specified frequency and is described as:

$$\Gamma_{i}(\omega) = \frac{\left(\Psi(e^{-i\omega})\sum \Psi'(e^{+i\omega})\right)_{ii}}{\frac{1}{2\pi}\int_{-\pi}^{\pi} \left(\Psi(e^{-i\lambda})\sum \Psi'(e^{+i\lambda})\right)_{ii} d\lambda}$$
(13)

The normalised generalised variance decomposition at the frequency band d can be calculated as follows:

$$\tilde{\theta}_{ij}(d) = \frac{\theta_{ij}(d)}{\sum_{j} \theta_{ij}(\infty)}$$
(14)

where  $\theta_{ij}(\infty)$  represents the contribution over all frequencies. All measures of connectedness are calculated as in (6)–(10) at a frequency band "d". These connectivity measures are established via a particular quantile  $\tau$ .

### 2.2. Data

To measure the spillover effect between investor sentiment and ESG investment volatility in Morocco, we use weekly data from the MASI ESG index, which was introduced in 2018. The goal of this index is to highlight companies performing well in terms of environmental, social, and governance factors relative to their market peers. The data are obtained from the Moroccan stock market website (https://www.casablanca-bourse.com/en/live-market/indices/ ESGI) and cover the period from January 2018 to December 2023. This period is instructive in terms of extreme events because it includes the COVID-19 outbreak and subsequent years, as well as the most recent geopolitical conflicts (The Russian-Ukrainian and Israeli-Palestinian conflicts).

The volatility of the ESG index is estimated on the basis of the conditional variance from the univariate GARCH model (Bollersley, 1986). In addition to the ESG index, we introduced the general market index, the Moroccan All Shares Index (MASI), to monitor general market conditions. The standard GARCH (p, q) model can be expressed as follows:

$$y_{t} = x_{t}\beta + \varepsilon_{t}, \ \varepsilon_{t} = h_{t}e_{t}, \ h_{t}^{2} = \phi + \sum_{i=1}^{p}\lambda_{i}\varepsilon_{t-i}^{2} + \sum_{j=1}^{q}\gamma_{j}h_{t-j}^{2}$$
(15)

where p represents the order of the moving average ARCH term, q denotes the order of the autoregressive GARCH term,  $y_t$  stands for the conditional mean,  $h_t^2$  expresses the conditional variance,  $\varepsilon_t^2$  is the squared residual,  $\lambda_i$  represents the ARCH parameters,  $\gamma_i$  represents the GARCH parameters,  $\dot{e_t}$  denotes a white noise process, and  $\phi$  is a constant.

We use the X (Twitter) sentiment as a proxy for investor sentiment. To construct this sentiment index, we applied the following procedure. We collected daily posts (tweets) related to "sustainable investment in Morocco" from the X platform for the period under investigation. In this context, we employed various related keywords, such as "Sustainable investment in Morocco", "Environmental, social, and governance criteria in Morocco", "Sustainability in Morocco", and "ESG in Morocco". Once we had cleaned the raw posts, we attributed a sentiment score to each post on the basis of natural language processing via Python. We obtained the opinion polarity from a total of 11,346 posts indicating whether a person has a neutral, positive or negative opinion toward sustainability. We then calculated the weekly average sentiment to construct a time-series of the weekly sentiment index with the following formula:

$$Tsent_{t} = \sum_{i=1}^{n} \frac{Sent_{it}}{N_{t}}$$
(16)

where  $Tsent_t$  represents the weekly sentiment at time t,  $Sent_{it}$  represents the sentiment score of message i posted at time t, and  $N_t$  denotes the number of messages posted at time t. The sentiment index ranges from -1 to 1, in which -1 is considered extremely pessimistic, 1 is considered extremely optimistic, and zero is considered neutral.

	Sentiment	ESG return	MASI return	ESG_Vol	MASI_Vol
Mean	0.106	-0.0004	-0.0001	0.016	0.015
Median	0.097	0.0013	0.0010	0.013	0.012
Max	0.480	0.0466	0.042	0.062	0.066
Min	-0.156	-0.0973	-0.09	0.009	0.009
Relative Std. dev.	1.037	48	179	0.521	0.549
Kurtosis	3.011	8.7355	10.144	13.130	18.038
Skewness	0.727	-1.4070	-1.6622	2.9781	3.557
JB	23.158***	549.68***	831.97***	1801.2***	3609.7***
ADF	-6.64***	-13.58***	-12.80***	-4.49**	-5.34***
ARCH–LM		56.18***	50.54***		

Table 1. Descriptive statistics of the series

Notes: JB represents the Jarque and Bera (1980) test for normality; ADF represents the augmented Dickey and Fuller unit root test; ARCH-LM represents Engle's (1982) ARCH-LM test. (\*\*\*) and (\*\*) denote significance at the 0.01 and 0.05 levels, respectively. Returns are calculated via the following formula:  $R_i = \ln(p_i/p_{i-1})$ , where  $p_i$  is the closing price of the index at time t and  $p_{t-1}$  is the closing price of the index at time t - 1.

Source: own elaboration.

Table 1 shows summary statistics for the sentiment, return, and volatility series. The ESG and MASI returns have a negative mean, indicating that the indices are losing value on average. However, the positive mean of the sentiment series implies that investors are, on average, optimistic about sustain-

able investment. Furthermore, we note that the relative standard deviation<sup>3</sup> of the sentiment series is lower than that of the ESG and MASI returns, indicating that sentiment is less volatile than returns. We also notice that the returns are negatively skewed due to their negative variations. The sentiment index is positively skewed, which means that the distribution is skewed to the right due to positive sentiment variations. The kurtosis is greater in all series than in the normal distribution. The distributions are therefore non-Gaussian. The nonnormality of all variables is also validated by the Jarque-Bera normality test. All these characteristics support the adoption of the *QVAR* connectedness approach.

To verify the stationarity of the variables, we employ the ADF test. All the series exhibit stationarity, as the ADF unit root test is highly significant. Finally, the significant ARCH-LM test suggests the presence of heteroscedasticity in the return series, supporting the use of the *GARCH* model in volatility estimation.



Figure 1. Dynamics of the investor sentiment index

Source: own elaboration.

Figure 1 illustrates the dynamics of the investor sentiment index. The results indicate that social media sentiment varies over time, ranging from -0.2 to 0.5. A negative value means that people are pessimistic about ESG investments, a positive value means that they are optimistic, and a zero means they are neutral. Specifically, we observe positive values most of the time, with important levels during crises, notably at the start of COVID-19, after the beginning of the war between Russia and Ukraine, and at the end of 2023,

<sup>&</sup>lt;sup>3</sup> The relative standard deviation stands for the standard deviation divided by the absolute mean.

which corresponds to the Israeli-Palestinian conflict. Nevertheless, during the COVID-19 pandemic, there were a few brief periods of negative sentiment, implying that during the crisis period, negative sentiment can occur due to significant negative shocks and uncertainty (Zribi et al., 2024). However, negative shocks in sentiment are short-lived, and people quickly switch to optimistic sentiments as the crisis raises awareness of the need for sustainability-related investments.

# 3. Results and discussion

In this section, we examine the static and dynamic connectedness between sentiment and ESG index volatility. Table 2 shows the static connectedness at three quantiles, namely, the median ( $\tau = 0.5$ ), the lower quantile ( $\tau = 0.05$ ), and the upper quantile ( $\tau = 0.95$ ).

The results show that the total connectedness index (*TCI*) is 37.42% at the median (normal market conditions), 41.06% at the lower quantile (bearish market conditions) and 64.10% at the upper quantile (bullish market conditions), which means that the *TCI* is higher under extreme quantiles than under the median, and that it is greater at the upper tail than at the lower tail. Furthermore, the findings show a significant feedback channel between ESG sentiment and volatility, particularly at the 95<sup>th</sup> quantile (where spillover effects range from 30.25% to 30.73%). This confirms that the risk in the upper tails is greater than that in the lower tails.

However, the static measurement of spillovers assumes that the coefficients of the model remain constant for the entire study period, which does not allow us to capture the structural breaks that occur because of extreme events. Moreover, numerous studies show that the impact of investor behaviour on financial markets evolves over time (El Oubani, 2022; Lekhal & El Oubani, 2020; Lo, 2004). To overcome this drawback, we re-estimate the time-varying connectedness between ESG sentiment and volatility.

Figure 2 shows the *TCI* in the time and frequency domains between investor sentiment toward sustainability, ESG index volatility, and MASI index volatility. The findings indicate that the *TCI* varies over time and depends on certain events. Indeed, notable peaks are observed in March 2020 (67%), February 2022 (66%), and October 2023 (60%), coinciding with the COVID-19 pandemic, the Russian-Ukrainian conflict, and the Israeli-Palestine conflict, respectively. Thus, during the crisis, the shocks created additional uncertainty, which was then transmitted to the entire system, leading to significant spillover into the system and thereby extremely high market risk. These results confirm those of Mensi et al. (2021), who argue that intense economic

	Sent.	ESG	MASI	FROM			
Panel A: Connectedness at the median							
Sent.	83.50	8.58	7.92	16.50			
ESG	2.35	52.83	44.82	47.17			
MASI	1.93	46.67	51.40	48.60			
то	4.28	55.25	52.74	112.27			
NET	-12.22	8.08	4.14	37.42			
Panel B: Connectedness at the 5 <sup>th</sup> quantile							
Sent.	79.05	11.08	9.88	20.95			
ESG	6.69	48.58	44.73	51.42			
MASI	6.02	44.80	49.18	50.82			
то	12.71	55.88	54.60	123.19			
NET	-8.24	4.46	3.78	41.06			
Panel C: Connectedness at the 95 <sup>th</sup> quantile							
Sent.	37.18	30.73	32.09	62.82			
ESG	30.25	34.60	35.15	65.40			
MASI	30.13	33.94	35.93	64.07			
то	60.37	64.67	67.24	192.29			
NET	-2.45	-0.73	3.18	64.10			

# Table 2. Sentiment and ESG volatility connectedness at the median, 5th quantileand 95th quantile

Notes: The variance decomposition is based on the *QVAR* approach, estimated at the median, lower quantile, and upper quantile, with a lag order of 1 chosen according to the Bayesian Information Criterion (BIC). The total connectedness index (*TCl*) in bold in the lower right corner is calculated via Eq. (6). The *TO, FROM*, and *NET* are calculated via Eq. (7), Eq. (8) and Eq. (9), respectively. The *ij*<sup>th</sup> value is the directional connectedness from variable *j* to variable *i* and is calculated via Eq. (3). The results are expressed as percentages.

Source: own elaboration.

periods create feelings of fear and uncertainty, which results in strong spillover effects. The results also support the findings of Liu et al. (2023) that exposure to COVID-19 increases stock market volatility. Nevertheless, they do not confirm some earlier studies suggesting that companies with better ESG scores have lower stock price volatility in times of crisis, such as COVID-19 (Albuquerque et al., 2020; Broadstock et al., 2021) or that ESG stocks have a safe-haven property (Rubbaniy et al., 2022).

A decline in spillover effects was observed during stable periods coinciding with the recovery of the economy from the COVID-19 epidemic (at the end



Figure 2. Short-term, long-term, and overall dynamic total connectedness

Notes: The results are estimated via the median QVAR model with a 52-week rolling window, a lag length of order 1 (BIC) and a generalised forecast error variance decomposition at 100 steps forward. The black area represents the time dynamic connectedness, the green area represents the long-term connectedness, and the blue area represents the short-term connectedness.

Source: own elaboration.

of 2021). This is due to lower uncertainty transmission, leading to low connectivity. Our results thus support Hypotheses 1 and 3.

To understand how the time horizon of investments can explain the origin of spillovers in the system, we investigate the frequency dynamics of connectivity. Market participants have heterogeneous expectations and therefore heterogeneous frequency responses to shocks. Ferrer et al. (2018) argue that diverse market participants have distinct goals, priorities, risk appetites and institutional constraints. While institutional investors prioritise long-term market returns, individual traders and hedge funds focus on short-term returns on assets, which results in heterogeneity in their frequency responses. Thus, asset prices, driven by fluctuations in demand involving multiple cyclical components, trigger shocks with different frequency responses, producing both short- and long-term spillovers.

The results in Figure 2 show that dynamic spillovers are induced by heterogeneous frequencies, in line with investors' heterogeneous perceptions and preferences. Nevertheless, spillovers are driven mainly by the long-term component, as they are concentrated at low frequencies, implying that the connectivity between ESG sentiment and volatility occurs mainly in the long term. These results are consistent with the fact that volatility connectedness is generally more persistent than return connectedness is (Wan et al., 2024; Wei et al., 2022), and that volatility often shows strong persistence, which is a stylised fact in financial markets. This characteristic was more pronounced during the COVID-19 epidemic and during the Russian-Ukrainian and Israeli-Palestinian conflicts. This can be explained by the fact that investors react very slowly to news, due to the important level of uncertainty during the crisis period, resulting in the shock being transmitted over a longer period and hence longer-term spillover effects. These results are consistent with those of Baruník and Křehlík (2018). They also align with the results of Wan et al. (2024), according to which attention to ESG has a significantly positive effect on the *TCI* of volatility in the low-frequency domain, suggesting that high attention to ESG results in strong volatility connectivity in the long term. These findings support Hypothesis 4.

To focus on the relationship between investor sentiment and ESG index volatility, we calculated the net pairwise spillover effect between these variables. The results in Figure 3 show a feedback channel between sentiment and ESG index volatility, in which investor sentiment toward sustainability acts as a net receiver of ESG volatility shocks most of the time, but becomes a net sender of shocks during certain periods. Interestingly, the net reception of shocks by investor sentiment is more pronounced during periods of extreme events, namely, the COVID-19 health crisis and geopolitical crises such as the Russian-Ukrainian and Israeli-Palestinian conflicts, suggesting that during these periods, shocks in the ESG index volatility prompt investors to pay



#### Figure 3. Short-term, long-term, and overall dynamic net pairwise connectedness between sentiment and ESG index volatility

Notes: The results are estimated via the median QVAR model with a 52-week rolling window, a lag length of order 1 (BIC) and a generalised forecast error variance decomposition at 100 steps forward. The black area depicts the time dynamic connectedness values, whereas the green and blue areas represent the long-term and short-term connectedness values, respectively.

Source: own elaboration.

more attention to what is being said about sustainable investment in social media. This finding is in line with that of Wan et al. (2024), who found that the level of attention given to ESG has risen sharply since 2020, implying that ESG attention has been strongly influenced by the pandemic. The high attention given to ESG investing can generate shocks in sentiment and therefore encourage more investors to include ESG stocks in their portfolios, leading to significant movements in ESG index returns, with sentiment thus becoming a net sender of shocks to ESG market volatility. This validates the conclusions reached by Li et al. (2023), who emphasised that investor panic caused by the COVID-19 epidemic is a possible trigger for risk spillovers in the stock market. Our findings also validate those of Liu et al. (2022), who found that ESG sentiment is positively related to the volatility risk premium, particularly the impact of environmental and social factors. Nevertheless, these results are not consistent with those of Dhasmana et al. (2023), who found that investor sentiment does not affect the ESG index in the Indian context, nor with those of Boubaker et al. (2022), who showed that responsible investment improved companies' resilience to the negative shocks of the COVID-19 health crisis. Therefore, our results confirm Hypothesis 2.

Like the total connectedness index, the net pairwise spillovers between ESG sentiment and volatility are also concentrated at low frequencies. This suggests that this interconnection is driven mainly by long-term fluctuations, indicating that markets do not react quickly to change. This observation is in line with the findings of El Oubani (2024), who found that the connectivity between ESG sentiment and volatility is due mainly to the long-term component. It also highlights the prevalence of strong risk aversion, an inherently unstable financial market, and a significant level of uncertainty, as confirmed by Baruník and Křehlík (2018). Together, these elements lead to slower shock transmission and more persistent investor reactions to these shocks. Consequently, these findings also validate Hypothesis 4.

To examine the dynamic connectedness between investor sentiment and ESG index volatility under different market conditions, we utilise the quantile connectedness approach of Ando et al. (2022). The empirical results in Figure 4 show that the *TCI* is stronger in the upper quantiles than in the other quantiles. This phenomenon can be attributed to investors' tendency to overreact to extreme shocks in the ESG market, as shown by previous studies (Dhasmana et al., 2023; Guo & Shi, 2024; Yang et al., 2024). Notable episodes include the COVID-19 health crisis in the first part of 2020 and the war between Russia and Ukraine in 2022, which caused disruptions in the supply of oil and other raw commodities. This means that periods of crisis are accompanied by high volatility and therefore high connectivity in the system, indicating high market risk. This contradicts some views that regard ESG stocks as safe havens (Rubbaniy et al., 2022). Importantly, investing in ESG stock indices does not protect against the risks that financial markets confront uniformly,



Figure 4. Quantile total connectedness between sentiment and volatility

Notes: The results are estimated via the median QVAR model with a 52-week rolling window, a lag length of order 1 (BIC) and a generalised forecast error variance decomposition at 100 steps forward. Warmer shades on the graph reflect greater levels of connectedness.

Source: own elaboration.

as their performance and volatility can be affected by a variety of extreme events (Chen & Lin, 2022). This encourages investors to implement effective risk management measures when investing in ESG markets, and to proactively hedge against potential shocks.

Now, we concentrate the analysis on the pairwise connectivity between sentiment toward sustainability and ESG volatility under different market conditions. On the basis of Figure 5, the estimation results show that investor



Figure 5. Quantile net pairwise connectedness between ESG sentiment and volatility

Notes: The results are estimated via the median QVAR model with a 52-week rolling window, a lag length of order 1 (BIC) and a generalised forecast error variance decomposition at 100 steps forward. Warmer shades indicate net-transmission.

Source: own elaboration.

sentiment transmits shocks to ESG index volatility mainly in the upper quantiles, implying that the impact of investor sentiment on ESG index volatility is more pronounced during positive extreme market conditions. In line with our previous results, the most notable events are the COVID-19 epidemic and the war between Russia and Ukraine, both of which are characterised by uncertainty and therefore high volatility. Indeed, media-generated panic during crisis periods is associated with increased market volatility (Haroon & Rizvi, 2020). These results are consistent with those of Naeem et al. (2020), who reported that investor sentiment plays a leading role in stock market volatility. These findings corroborate Hypothesis 5 and suggest that sentiment could be used to predict ESG volatility in the Moroccan stock market, particularly in the upper quantiles, to achieve abnormal returns. Consequently, investors need to make timely adjustments to their ESG portfolios on the basis of investor sentiment. Policymakers also need to consider extreme positive tails when making regulatory decisions.

# 4. Robustness tests

We complete our analysis by conducting some robustness tests on our results. First, we focused our analysis on social media sentiment to capture investors' attention to sustainability. However, this could be biased in terms of representing a broader investor population. To diversify the data sources and confirm the robustness of our results, we use online search attention as an alternative to social media sentiment. Second, we consider whether our findings differ if we change certain parameters of the *QVAR* method. The analyses are repeated to check whether the same results are obtained.

## 4.1. Online search attention and ESG volatility

Following Da et al. (2011) and Duc et al. (2024), we employ the Google Volume Search Index (*GVSI*) using sustainability-related terms obtained via Google Trends, to assess investors' attention to ESG investing. In fact, the *GSVI* for the topic "Environmental, social and governance criteria"<sup>4</sup> in Morocco is used. Next, an abnormal *GSVI* (*AGSVI*) is calculated via the following formula:

<sup>&</sup>lt;sup>4</sup> To check the robustness of the results, other keywords closely related to sustainable investment were used. However, the results for these keywords were not significant.

$$AGSVI_{t} = \frac{Weekly\,GSVI_{t}}{Mean_{t}^{24}} \tag{17}$$

in which  $Mean_t^{24}$  is the 24-week moving average before time t.

We then test the dynamic interplay between investor attention to ESG, captured by online search intensity, and ESG volatility via the *QVAR* approach. The results reported in Figure 6 are consistent with those observed when we consider social media sentiment. This confirms the relevance of using social media sentiment to track people's attitudes toward sustainability.

#### 4.2. Changes in the QVAR parameters

To check the robustness of the method used in this study, we re-estimate the connectedness between ESG sentiment and volatility via the *QVAR* with the modification of certain parameters such as the forecast horizon, the rolling window size, and the model selection criterion. The results presented in Figure 7 show consistency between the results when the forecast horizon, the rolling window size, and the model selection criterion are modified in the *QVAR* method, implying that the choice of these parameters does not significantly affect the conclusions of the baseline study.

# 5. Study implications

The outcomes of the study are beneficial for pro-ESG investors, companies, and policymakers. The results suggest that investors should consider what is posted on social media when investing in the ESG market to strategically manage their portfolios, as social media sentiment influences returns and risks in this market. This can help investors build profitable investment strategies, such as momentum strategies (Lekhal & El Oubani, 2020). Indeed, investors systematically exaggerate ESG information disclosed by companies, leading to ESG momentum effects in financial markets that can translate into substantial short-term profits (Chen & Yang, 2020). Additionally, the results help investors manage their ESG portfolios in times of crisis by considering arbitrage opportunities. In fact, we find that sentiment amplified long-term risk in the ESG market during the COVID-19 pandemic because sentiment transmitted persistent shocks to volatility. This finding is confirmed in the US, Europe and emerging markets, where the panic index induced by the COVID-19 pandemic had a significant effect on ESG volatility (Umar et al., 2021). However, this finding does





Notes: The results are estimated via the QVAR model with a 52-week rolling window, a lag length of order 1 (BIC) and a generalised forecast error variance decomposition at 100 steps forward. In *a* and *b*, the black area depicts the time dynamic connectedness, whereas the green and blue areas represent the long-term and short-term connectedness, respectively. Warmer shades reflect greater levels of connectedness in *c*. Warmer shades indicate net-transmission in *d*.

Source: own elaboration.





Notes: The results are estimated via the *QVAR* model with a 100-week rolling-window size, a lag length of order 1 (AIC), and a generalised forecast error variance decomposition at 200 steps forward. In *a* and *b*, the black area depicts the time dynamic connectedness, whereas the green and blue areas represent the long-term and short-term connectedness, respectively. Warmer shades reflect greater levels of connectedness in *c*. Warmer shades indicate net-transmission in *d*.

Source: own elaboration.

not agree with the findings of Liu et al. (2023), according to which strong ESG performance contributed to enhancing stock market stability in Japan during the COVID-19 pandemic. Consequently, the extent to which sentiment influences decision-making and therefore market risk depends on cultural differences. This enables investors to take advantage of cultural differences by using arbitrage strategies. In this context, Naeem et al. (2023) found that the COVID-19 outbreak generated arbitrage opportunities in the ESG markets of the US, Latin America and Asia-Pacific region, but not in Europe. It is therefore important to consider the specific characteristics of each market when investing in ESG indices. Furthermore, our results enable investors to detect under which market conditions sentiment can strongly predict movements in ESG markets. We find that in times of health or geopolitical crisis, the spillover effects between sentiment and volatility are greater in the upper quantiles than in the other quantiles, suggesting that profitable trading strategies can be conceived in such times to exploit the large price movements triggered by sentiment.

Furthermore, our results suggest that companies should consider sentiment when communicating their ESG performance. In fact, financial performance depends not only on a company's ESG score, but also on investor sentiment, particularly during turbulent events. Specifically, the findings show that sentiment influences volatility, notably during the COVID-19 health crisis and geopolitical crises such as the Russian-Ukrainian and Israeli-Palestinian conflicts. This could be explained by the fact that uncertainties prevailing in such periods make investors more sensitive to environmental and social risks, leading to a shift in investor attention from traditional investing to ESG investing and therefore to greater trading activity in the ESG market, which could cause significant fluctuations in stock prices. However, this greater attention might make it easier to raise funds in the market.

As far as regulators are concerned, the results suggest that they should strengthen communication on sustainability guidance, and use social media to draw investors' attention to ESG, particularly during periods of crisis such as the COVID-19 pandemic. This could be a signal to the government to encourage a green recovery in times of crisis, promoting ESG practices and thus fostering sustainable economic growth (Liu et al., 2023). In other words, the government should focus on leveraging the pandemic crisis to advance climate activism, develop a circular economy, and move toward resilient social and economic models capable of resisting disasters and infectious diseases (Taghizadeh-Hesary & Rasoulinezhad, 2023), thereby achieving UN's objectives. However, the strong connectedness observed between sentiment and volatility, particularly during extreme events, increases risk in the ESG market and makes it inefficient and unstable. In this context, policymakers should implement measures and regulations that encourage investors to undertake long-term investment instead of engaging in speculation that causes market instability and inefficiency. In addition, the implementation of standard

frameworks and guidelines for ESG reporting might improve transparency and comparability between companies, providing investors with trustworthy information on companies' environmental and social performance, thereby facilitating better decision-making and lowering market volatility (Liu et al., 2023). In fact, ESG commitments contribute to strengthening public confidence in a company, providing the prospect of significant protection against losses in times of crisis. Lins et al. (2017) asserted that ESG advantages are not constant over time, but are linked to the general level of confidence in companies and financial markets, and that these advantages tend to be greater when confidence is at its lowest. Furthermore, stock exchanges need to strengthen ESG disclosure requirements to enable investors to make informed decisions and reduce the market uncertainty that prevails in times of crisis.

# Conclusions

The aim of this study is to examine the dynamic interplay between investor sentiment toward sustainable investment and the volatility of the ESG index in the Moroccan financial market. To this end, we constructed an investor sentiment index from the X platform (Twitter) and used GARCH-based ESG index volatility derived from weekly returns, as well as guantile and frequency connectedness approaches. The results show that the spillover effects between investor sentiment and ESG index volatility evolve over time, depending on distress events like the COVID-19 health crisis and geopolitical crises. Moreover, most of the time sentiment acts as a net receiver of shocks from ESG market volatility, mainly in times of crisis, but sometimes becomes a net sender of shocks. Furthermore, the results reveal that long-term spillovers dominate short-term spillovers, particularly during extreme events, implying the persistence of shock transmission due to high uncertainty. Finally, we observe that connectedness is stronger in the upper quantiles than in the middle or lower quantiles, suggesting the impact of extreme positive market conditions on spillover effects between ESG sentiment and volatility.

We investigated the connectedness between investor sentiment and the volatility of the ESG index in Morocco, but it would also be interesting to examine this connectedness across different developed and emerging countries. Additionally, it would be highly representative to use different ESG sentiment proxies, or to construct an aggregate ESG sentiment index combining the most relevant sentiment indicators. Moreover, it would be particularly useful to measure how sentiment expressed in social media improves the predictability of returns and the volatility of the ESG market. These questions could provide important directions for future research.

# References

- Abdollahi, H., Fjesme, S. L., & Sirnes, E. (2024). Measuring market volatility connectedness to media sentiment. *The North American Journal of Economics and Finance*, 71, 102091. https://doi.org/10.1016/j.najef.2024.102091
- AlZaabi, S. A. (2021). Correlating sentiment in Reddit's Wallstreetbets with the stock market using machine learning techniques. Rochester Institute of Technology [thesis].
- Albuquerque, R., Koskinen, Y., Yang, S., & Zhang, C. (2020). Resiliency of environmental and social stocks: An analysis of the exogenous COVID-19 market crash. *The Review* of Corporate Finance Studies, 9(3), 593–621. https://doi.org/10.1093/rcfs/cfaa011
- Ando, T., Greenwood-Nimmo, M., & Shin, Y. (2022). Quantile connectedness: Modeling tail behavior in the topology of financial networks. *Management Science*, 68(4), 2401–2431. https://doi.org/10.1287/mnsc.2021.3984
- Bannier, C. E., Bofinger, Y., & Rock, B. (2022). Corporate social responsibility and credit risk. *Finance Research Letters*, 44, 102052. https://doi.org/10.1016/ j.frl.2021.102052
- Baruník, J., & Křehlík, T. (2018). Measuring the frequency dynamics of financial connectedness and systemic risk. *Journal of Financial Econometrics*, 16(2), 271–296. https://doi.org/10.1093/jjfinec/nby001
- Bofinger, Y., Heyden, K. J., & Rock, B. (2022). Corporate social responsibility and market efficiency: Evidence from ESG and misvaluation measures. *Journal of Banking* & Finance, 134, 106322. https://doi.org/10.1016/j.jbankfin.2021.106322

Bollerslev, T. (1986). Generalized autoregressive conditional heteroskedasticity. *Journal* of *Econometrics*, *31*(3), 307–327. https://doi.org/10.1016/0304-4076(86)90063-1

- Boubaker, S., Liu, Z., & Zhan, Y. (2022). Customer relationships, corporate social responsibility, and stock price reaction: Lessons from China during health crisis times. *Finance Research Letters*, 47, 102699. https://doi.org/10.1016/j.frl.2022.102699
- Broadstock, D. C., Chan, K., Cheng, L. T., & Wang, X. (2021). The role of ESG performance during times of financial crisis: Evidence from COVID-19 in China. *Finance Research Letters*, 38, 101716. https://doi.org/10.1016/j.frl.2020.101716
- Caldara, D., & Iacoviello, M. (2022). Measuring geopolitical risk. *American Economic Review*, *112*(4), 1194–1225. https://doi.org/10.1257/aer.20191823
- Chen, H. Y., & Yang, S. S. (2020). Do investors exaggerate corporate ESG information? Evidence of the ESG momentum effect in the Taiwanese market. *Pacific-Basin Finance Journal*, *63*, 101407. https://doi.org/10.1016/j.pacfin.2020.101407
- Chen, Y., & Lin, B. (2022). Quantifying the extreme spillovers on worldwide ESG leaders' equity. *International Review of Financial Analysis*, *84*, 102425. https://doi. org/10.1016/j.irfa.2022.102425
- Cheng, R., Kim, H., & Ryu, D. (2024). ESG performance and firm value in the Chinese market. *Investment Analysts Journal*, *53*(1), 1–15. https://doi.org/10.1080/1029 3523.2023.2218124
- Chiu, C. W. J., Harris, R. D., Stoja, E., & Chin, M. (2018). Financial market volatility, macroeconomic fundamentals and investor sentiment. *Journal of Banking & Finance*, 92, 130–145. https://doi.org/10.1016/j.jbankfin.2018.05.003

- Cunado, J., Chatziantoniou, I., Gabauer, D., de Gracia, F. P., & Hardik, M. (2023). Dynamic spillovers across precious metals and oil realized volatilities: Evidence from quantile extended joint connectedness measures. *Journal of Commodity Markets*, *30*, 100327. https://doi.org/10.1016/j.jcomm.2023.100327
- Da, Z., Engelberg, J., & Gao, P. (2011). In search of attention. *The Journal of Finance*, *66*(5), 1461–1499. https://doi.org/10.1111/j.1540-6261.2011.01679.x
- Daugaard, D. (2020). Emerging new themes in environmental, social and governance investing: A systematic literature review. *Accounting & Finance*, *60*(2), 1501–1530. https://doi.org/10.1111/acfi.12479
- Demers, E., Hendrikse, J., Joos, P., & Lev, B. (2021). ESG did not immunize stocks during the COVID-19 crisis, but investments in intangible assets did. *Journal of Business Finance & Accounting*, 48(3–4), 433–462. https://doi.org/10.1111/jbfa.12523
- Dhasmana, S., Ghosh, S., & Kanjilal, K. (2023). Does investor sentiment influence ESG stock performance? Evidence from India. *Journal of Behavioral and Experimental Finance*, *37*, 100789. https://doi.org/10.1016/j.jbef.2023.100789
- Diebold, F. X., & Yilmaz, K. (2012). Better to give than to receive: Predictive directional measurement of volatility spillovers. *International Journal of Forecasting*, 28(1), 57–66. https://doi.org/10.1016/j.ijforecast.2011.02.006
- Döttling, R., & Kim, S. (2024). Sustainability preferences under stress: Evidence from COVID-19. *Journal of Financial and Quantitative Analysis*, *59*(2), 435–473. https://doi.org/10.1017/S0022109022001296
- Duc, D. T. V., Hoai, N. T., Nguyen, D. P., Anh, N. H., & Hai, H. H. (2024). Google Search intensity and stock returns in frontier markets: Evidence from the Vietnamese market. *Economics and Business Review*, 10(1), 30–56. https://doi.org/10.18559/ ebr.2024.1.778
- El Khoury, R., Alshater, M. M., Li, Y., & Xiong, X. (2024). Quantile time-frequency connectedness among G7 stock markets and clean energy markets. *The Quarterly Review of Economics and Finance*, *93*, 7190. https://doi.org/10.1016/j.qref.2023.11.004
- El Ouadghiri, I., Guesmi, K., Peillex, J., & Ziegler, A. (2021). Public attention to environmental issues and stock market returns. *Ecological Economics*, *180*, 106836. https://doi.org/10.1016/j.ecolecon.2020.106836
- El Oubani, A. (2022). Market efficiency dynamics during the COVID-19 pandemic. *Revue des Etudes Multidisciplinaires en Sciences Economiques et Sociales, 7*(2), 357–376. https://doi.org/10.48375/IMIST.PRSM/remses-v7i2.32041
- El Oubani, A. (2023). Complex adaptive behavior of investors to market dynamics: A PLS-SEM analysis. *Journal of Applied Structural Equation Modeling*, 7(2), 1–25. https://doi.org/10.47263/JASEM.7(2)03
- El Oubani, A. (2024). Investor sentiment and sustainable investment: evidence from North African stock markets. *Future Business Journal, 10*(1), 68. https://doi. org/10.1186/s43093-024-00349-x
- El Oubani, A., & Lekhal, M. (2022). An agent-based model of financial market efficiency dynamics. *Borsa Istanbul Review*, 22(4), 699–710. https://doi.org/10.1016/ j.bir.2021.10.005

- Engelhardt, N., Ekkenga, J., & Posch, P. (2021). ESG ratings and stock performance during the COVID-19 crisis. *Sustainability*, 13(13), 7133. https://doi.org/10.3390/ su13137133
- Engle, R. F. (1982). Autoregressive conditional heteroscedasticity with estimates of the variance of UK inflation. *Econometrica*, 50, 987–1008. https://doi.org/10. 2307/1912773
- Fama, E. F. (1970). Efficient capital markets: A review of theory an empirical work. *Journal of Finance*, *25*(2), 383–417. https://doi.org/10.2307/2325486
- Ferrer, R., Shahzad, S. J. H., López, R., & Jareño, F. (2018). Time and frequency dynamics of connectedness between renewable energy stocks and crude oil prices. *Energy Economics*, 76, 1–20. https://doi.org/10.1016/j.eneco.2018.09.022
- Ford, J. M., Gehricke, S. A., & Zhang, J. E. (2022). Option traders are concerned about climate risks: ESG ratings and short-term sentiment. *Journal of Behavioral and Experimental Finance*, 35, 100687. https://doi.org/10.1016/j.jbef.2022.100687
- Gao, Y., Li, Y., Zhao, C., & Wang, Y. (2022). Risk spillover analysis across worldwide ESG stock markets: New evidence from the frequency-domain. *The North American Journal of Economics and Finance*, 59, 101619. https://doi.org/10.1016/j.najef.2021.101619
- Garel, A., & Petit-Romec, A. (2021). Investor rewards to environmental responsibility: Evidence from the COVID-19 crisis. *Journal of Corporate Finance, 68*, 101948. https://doi.org/10.1016/j.jcorpfin.2021.101948
- Giannarakis, G., Partalidou, X., Zafeiriou, E., & Sariannidis, N. (2016). An analysis of United States on Dow Jones sustainability index. *Investment Management and Financial Innovations* 13(3), 353–361. http://dx.doi.org/10.21511/imfi.13(3-2). 2016.07
- Guo, P., & Shi, J. (2024). Geopolitical risks, investor sentiment and industry stock market volatility in China: Evidence from a quantile regression approach. *The North American Journal of Economics and Finance*, 72, 102139. https://doi.org/10.1016/ j.najef.2024.102139
- Haroon, O., & Rizvi, S. A. R. (2020). COVID-19: Media coverage and financial markets behavior—A sectoral inquiry. *Journal of Behavioral and Experimental Finance*, 27, 100343. https://doi.org/10.1016/j.jbef.2020.100343
- Hartzmark, S. M., & Sussman, A. B. (2019). Do investors value sustainability? A natural experiment examining ranking and fund flows. *The Journal of Finance*, 74(6), 2789–2837. https://doi.org/10.1111/jofi.12841
- Isaak, M., & Lentz, W. (2020). Consumer preferences for sustainability in food and nonfood horticulture production. *Sustainability*, 12(17), 7004. https://doi.org/10.3390/ su12177004
- Jarque, C. M., & Bera, A. K. (1980). Efficient tests for normality, homoscedasticity and serial independence of regression residuals. *Economics Letters*, *6*(3), 255–259. https://doi.org/10.1016/0165-1765(80)90024-5
- Khan, M. A., Khan, A., Hassan, M. K., & Maraghini, M. P. (2024). Market response to environmental social and governance performance: A global analysis. *Research in International Business and Finance*, 67, 102131. https://doi.org/10.1016/ j.ribaf.2023.102131

- Klink, S., & Gonciarenko, S. (2017, April 19). Philips couples sustainability performance to interest rate of its new EUR 1 billion Revolving Credit Facility [Press release]. https://www.philips.com/a-w/about/news/archive/standard/news/ press/2017/20170419-philips-couples-sustainability-performance-to-interestrate-of-its-new-eur-1-billion-revolving-credit-facility.html
- Koop, G., Pesaran, M. H., & Potter, S. M. (1996). Impulse response analysis in nonlinear multivariate models. *Journal of Econometrics*, 74(1), 119–147. https://doi. org/10.1016/0304-4076(95)01753-4
- Lei, H., Xue, M., Liu, H., & Ye, J. (2023). Precious metal as a safe haven for global ESG stocks: Portfolio implications for socially responsible investing. *Resources Policy*, 80, 103170. https://doi.org/10.1016/j.resourpol.2022.103170
- Lekhal, M., & El Oubani, A. (2020). Does the adaptive market hypothesis explain the evolution of emerging markets efficiency? Evidence from the Moroccan financial market. *Heliyon*, 6(7), e04429. https://doi.org/10.1016/j.heliyon.2020.e04429
- Li, H., Zhang, X., & Zhao, Y. (2022). ESG and firm's default risk. *Finance Research Letters*, 47, 102713. https://doi.org/10.1016/j.frl.2022.102713
- Li, Y., Shi, Y., Shi, Y., Yi, S., & Zhang, W. (2023). COVID-19 vaccinations and risk spillovers: Evidence from Asia-Pacific stock markets. *Pacific-Basin Finance Journal*, 79, 102004. https://doi.org/10.1016/j.pacfin.2023.102004
- Liao, Y., & Pan, Z. (2022). Extreme risk connectedness among global major financial institutions: Links to globalization and emerging market fear. *Pacific-Basin Finance Journal*, *76*, 101862. https://doi.org/10.1016/j.pacfin.2022.101862
- Lins, K. V., Servaes, H., & Tamayo, A. (2017). Social capital, trust, and firm performance: The value of corporate social responsibility during the financial crisis. *The Journal* of Finance, 72(4), 1785–1824. https://doi.org/10.1111/jofi.12505
- Liu, L., Nemoto, N., & Lu, C. (2023). The Effect of ESG performance on the stock market during the COVID-19 Pandemic—Evidence from Japan. *Economic Analysis and Policy*, 79, 702–712. https://doi.org/10.1016/j.eap.2023.06.038
- Liu, Z., Wang, S., Liu, S., Yu, H., & Wang, H. (2022). Volatility risk premium, return predictability, and ESG sentiment: Evidence from China's spots and options' markets. *Complexity*, (1), 6813797. https://doi.org/10.1155/2022/6813797
- Lo, A. W. (2004). The adaptive markets hypothesis. *The Journal of Portfolio Management*, 30(5), 15–29. https://doi.org/10.3905/jpm.2004.442611
- Londono, J. M. (2019). Bad bad contagion. *Journal of Banking & Finance, 108*, 105652. https://doi.org/10.1016/j.jbankfin.2019.105652
- López-Cabarcos, M. Á., Pérez-Pico, A. M., & López-Pérez, M. L. (2019). Does social network sentiment influence S&P 500 Environmental & Socially Responsible Index? *Sustainability*, 11(2), 320. https://doi.org/10.3390/su11020320
- Mensi, W., Al-Yahyaee, K. H., Vo, X. V., & Kang, S. H. (2021). Modeling the frequency dynamics of spillovers and connectedness between crude oil and MENA stock markets with portfolio implications. *Economic Analysis and Policy*, *71*, 397–419. https://doi.org/10.1016/j.eap.2021.06.001
- Murata, R., & Hamori, S. (2021). ESG disclosures and stock price crash risk. *Journal of Risk and Financial Management*, 14(2), 70. https://doi.org/10.3390/jrfm14020070

- Naeem, M. A., Farid, S., Faruk, B., & Shahzad, S. J. H. (2020). Can happiness predict future volatility in stock markets? *Research in International Business and Finance*, 54, 101298. https://doi.org/10.1016/j.ribaf.2020.101298
- Naeem, M. A., Yousaf, I., Karim, S., Tiwari, A. K., & Farid, S. (2023). Comparing asymmetric price efficiency in regional ESG markets before and during COVID-19. *Economic Modelling*, 118, 106095. https://doi.org/10.1016/j.econmod.2022.106095
- Nofsinger, J. R., Sulaeman, J., & Varma, A. (2019). Institutional investors and corporate social responsibility. *Journal of Corporate Finance*, *58*, 700–725. https://doi.org/10.1016/j.jcorpfin.2019.07.012
- Nofsinger, J., & Varma, A. (2014). Socially responsible funds and market crises. *Journal of Banking & Finance*, 48, 180–193. https://doi.org/10.1016/j.jbankfin.2013.12.016
- Nyakurukwa, K., & Seetharam, Y. (2023a). Investor reaction to ESG news sentiment: Evidence from South Africa. *Economia*, 24(1), 68–85. https://doi.org/10.1108/ ECON-09-2022-0126
- Nyakurukwa, K., & Seetharam, Y. (2023b). The evolution of studies on social media sentiment in the stock market: Insights from bibliometric analysis. *Scientific African*, *20*, e01596. https://doi.org/10.1016/j.sciaf.2023.e01596
- Nyakurukwa, K., & Seetharam, Y. (2024). Sentimental showdown: News media vs. social media in stock markets. *Heliyon*, *10*(9). e30211. https://doi.org/10.1016/ j.heliyon.2024.e30211
- Ortas, E., Moneva, J. M., & Salvador, M. (2014). Do social and environmental screens influence ethical portfolio performance? Evidence from Europe. *BRQ Business Research Quarterly*, *17*(1), 11–21. https://doi.org/10.1016/j.cede.2012.11.001
- Paramanik, R. N., & Singhal, V. (2020). Sentiment analysis of Indian stock market volatility. *Procedia Computer Science*, 176, 330–338. https://doi.org/10.1016/ j.procs.2020.08.035
- Pedini, L., & Severini, S. (2022). Exploring the hedge, diversifier and safe haven properties of ESG investments: A cross-quantilogram analysis. MPRA Paper, 112339. University Library of Munich, Germany. https://ideas.repec.org/p/pra/ mprapa/112339.html
- Pesaran, H. H., & Shin, Y. (1998). Generalized impulse response analysis in linear multivariate models. *Economics Letters*, 58(1), 17–29. https://doi.org/10.1016/ S0165-1765(97)00214-0
- PH, H., & Rishad, A. (2020). An empirical examination of investor sentiment and stock market volatility: Evidence from India. *Financial Innovation*, *6*(1), 1–15. https://doi.org/10.1186/s40854-020-00198-x
- Pitoska, E., Katarachia, A., Giannarakis, G., & Tsilikas, C. (2017). An analysis of determinants affecting the returns of Dow Jones sustainability index United States. *International Journal of Economics and Financial Issues*, 7(3), 113–118. https:// dergipark.org.tr/en/pub/ijefi/issue/32021/354207
- Raimo, N., Caragnano, A., Zito, M., Vitolla, F., & Mariani, M. (2021). Extending the benefits of ESG disclosure: The effect on the cost of debt financing. *Corporate Social Responsibility and Environmental Management*, 28(4), 1412–1421. https:// doi.org/10.1002/csr.2134

- Ren, X., Cao, Y., Liu, P. J., & Han, D. (2023). Does geopolitical risk affect firms' idiosyncratic volatility? Evidence from China. *International Review of Financial Analysis*, 90, 102843. https://doi.org/10.1016/j.irfa.2023.102843
- Rubbaniy, G., Khalid, A. A., Rizwan, M. F., & Ali, S. (2022). Are ESG stocks safe-haven during COVID-19? *Studies in Economics and Finance*, *39*(2), 239–255. https://doi. org/10.1108/SEF-08-2021-0320
- Sabbaghi, O. (2022). The impact of news on the volatility of ESG firms. *Global Finance Journal*, *51*, 100570. https://doi.org/10.1016/j.gfj.2020.100570
- Shakil, M. H. (2021). Environmental, social and governance performance and financial risk: Moderating role of ESG controversies and board gender diversity. *Resources Policy*, *72*, 102144. https://doi.org/10.1016/j.resourpol.2021.102144
- Sheikh, U. A., Asadi, M., Roubaud, D., & Hammoudeh, S. (2024). Global uncertainties and Australian financial markets: Quantile time-frequency connectedness. *International Review of Financial Analysis*, 92, 103098. https://doi.org/10.1016/ j.irfa.2024.103098
- Shi, Y., & Ho, K. Y. (2015). Long memory and regime switching: A simulation study on the Markov regime-switching ARFIMA model. *Journal of Banking & Finance*, 61, S189–S204. https://doi.org/10.1016/j.jbankfin.2015.08.025
- Shiller, R. J. (1981). Do stock prices move too much to be justified by subsequent changes in dividends? *The American Economic Review*, *71*(3), 421–436. https://www.aeaweb.org/aer/top20/71.3.421-436.pdf?mod=article\_inline
- Shutes, K., McGrath, K., Lis, P., & Riegler, R. (2016). Twitter and the US stock market: The influence of microbloggers on share prices. *Economics and Business Review*, 2(3), 57–77. https://doi.org/10.18559/ebr.2016.3.5
- Siganos, A., Vagenas-Nanos, E., & Verwijmeren, P. (2017). Divergence of sentiment and stock market trading. *Journal of Banking & Finance*, 78, 130–141. https://doi. org/10.1016/j.jbankfin.2017.02.005
- Simon, H. A. (1982). *Models of bounded rationality and other topics in economics* (vol. 2). MIT Press.
- Sohag, K., Hammoudeh, S., Elsayed, A. H., Mariev, O., & Safonova, Y. (2022). Do geopolitical events transmit opportunity or threat to green markets? Decomposed measures of geopolitical risks. *Energy Economics*, 111, 106068. https://doi. org/10.1016/j.eneco.2022.106068
- Suleman, M. T., Tabash, M. I., & Sheikh, U. A. (2024). Do stock market fluctuations lead to currency deflation in the South Asian region? Evidence beyond symmetry. *International Journal of Finance & Economics*, 29(2), 1432–1450. https://doi. org/10.1002/ijfe.2746
- Taghizadeh-Hesary, F., & Rasoulinezhad, E. (2023). 97 An empirical assessment of ways to achieve a green economic recovery in the post-pandemic era in ASEAN. In F. Taghizadeh-Hesary, N. Yoshino, N. Panthamit, H. Phoumin (Eds.), *Post-pandemic green recovery in ASEAN* (pp. 97–116). Routledge.
- Tversky, A., & Kahneman, D. (1974). Judgment under Uncertainty: Heuristics and Biases: Biases in judgments reveal some heuristics of thinking under uncertainty. *Science*, 185(4157), 1124–1131. https://doi.org/10.1126/science.185.4157.1124
- Umar, Z., Gubareva, M., Tran, D. K., & Teplova, T. (2021). Impact of the COVID-19 induced panic on the Environmental, Social and Governance leaders equity volatili-

ty: A time-frequency analysis. *Research in International Business and Finance, 58,* 101493. https://doi.org/10.1016/j.ribaf.2021.101493

- Vuong, N. B. (2022). Investor sentiment, corporate social responsibility, and financial performance: Evidence from Japanese companies. *Borsa Istanbul Review*, 22(5), 911–924. https://doi.org/10.1016/j.bir.2022.06.010
- Wan, J., Yin, L., & Wu, Y. (2024). Return and volatility connectedness across global ESG stock indexes: Evidence from the time-frequency domain analysis. *International Review of Economics & Finance*, 89, 397–428. https://doi.org/10.1016/ j.iref.2023.10.038
- Wang, H., Xu, L., & Sharma, S. S. (2021). Does investor attention increase stock market volatility during the COVID-19 pandemic? *Pacific-Basin Finance Journal*, 69, 101638. https://doi.org/10.1016/j.pacfin.2021.101638
- Wang, W., Sun, Z., Wang, W., Hua, Q., & Wu, F. (2023). The impact of environmental uncertainty on ESG performance: Emotional vs. rational. *Journal of Cleaner Production*, *397*, 136528. https://doi.org/10.1016/j.jclepro.2023.136528
- Wei, Y., Zhang, Y., & Wang, Y. (2022). Information connectedness of international crude oil futures: Evidence from SC, WTI, and Brent. *International Review of Financial Analysis*, 81, 102100. https://doi.org/10.1016/j.irfa.2022.102100
- Willett, T. D. (2024). Some implications of behavioral finance for international monetary analysis. *Economics and Business Review*, *10*(1), 7–29. https://doi. org/10.18559/ebr.2024.1.1193
- Wu, Y., Ren, W., Wan, J., & Liu, X. (2023). Time-frequency volatility connectedness between fossil energy and agricultural commodities: Comparing the COVID-19 pandemic with the Russia-Ukraine conflict. *Finance Research Letters*, 55, 103866. https://doi.org/10.1016/j.frl.2023.103866
- Yang, J., Agyei, S. K., Bossman, A., Gubareva, M., & Marfo-Yiadom, E. (2024). Energy, metals, market uncertainties, and ESG stocks: Analysing predictability and safe havens. *The North American Journal of Economics and Finance*, 69, 102030. https:// doi.org/10.1016/j.najef.2023.102030
- Yu, H., Liang, C., Liu, Z., & Wang, H. (2023). News-based ESG sentiment and stock price crash risk. *International Review of Financial Analysis*, 88, 102646. https:// doi.org/10.1016/j.irfa.2023.102646
- Zhan, J. X., & Santos-Paulino, A. U. (2021). Investing in the Sustainable Development Goals: Mobilization, channeling, and impact. *Journal of International Business Policy*, 4(1), 166–183. https://doi.org/10.1057%2Fs42214-020-00093-3
- Zhang, A. Y., & Zhang, J. H. (2023). Renovation in environmental, social and governance (ESG) research: The application of machine learning. *Asian Review of Accounting*. https://doi.org/10.1108/ARA-07-2023-0201
- Zribi, W., Boufateh, T., Lahouel, B. B., & Urom, C. (2024). Uncertainty shocks, investor sentiment and environmental performance: Novel evidence from a PVAR approach. *International Review of Financial Analysis*, 93, 103196. https://doi.org/10.1016/ j.irfa.2024.103196

#### Aims and Scope

The **Economics and Business Review** is a quarterly journal focusing on theoretical, empirical and applied research in the fields of Economics and Corporate and Public Finance. The Journal welcomes the submission of high quality articles dealing with micro, mezzo and macro issues well founded in modern theories and relevant to an international audience. The EBR's goal is to provide a platform for academicians all over the world to share, discuss and integrate state-of-the-art Economics and Finance thinking with special focus on new market economies.

#### The manuscript

- Articles submitted for publication in the Economics and Business Review should contain original, unpublished work not submitted for publication elsewhere.
- 2. Manuscripts intended for publication should be written in English, edited in Word in accordance with the APA editorial guidelines and sent to: secretary@ebr.edu.pl. Authors should upload two versions of their manuscript. One should be a complete text, while in the second all document information identifying the author(s) should be removed from papers to allow them to be sent to anonymous referees.
- 3. Manuscripts are to be typewritten in **12' font in A4 paper** format, one and half spaced and be aligned. Pages should be numbered. Maximum size of the paper should be up to 20 pages.
- 4. Papers should have an abstract of about 100-150 words, keywords and the Journal of Economic Literature classification code (JEL Codes).
- 5. Authors should clearly declare the aim(s) of the paper. Papers should be divided into numbered (in Arabic numerals) sections.
- 6. Acknowledgements and references to grants, affiliations, postal and e-mail addresses, etc. should appear as a separate footnote to the author's name a, b, etc and should not be included in the main list of footnotes.
- 7. Footnotes should be listed consecutively throughout the text in Arabic numerals. Cross-references should refer to particular section numbers: e.g.: See Section 1.4.
- 8. Quoted texts of more than 40 words should be separated from the main body by a four-spaced indentation of the margin as a block.
- References The EBR 2022 editorial style is based on the 7th edition of the Publication Manual of the American Psychological Association (APA). For more information see APA Style used in EBR guidelines.
- 10. **Copyrights** will be established in the name of the **EBR publisher**, namely the Poznań University of Economics and Business Press.

More information and advice on the suitability and formats of manuscripts can be obtained from:

Economics and Business Review al. Niepodległości 10 61-875 Poznań Poland e-mail: secretary@ebr.edu.pl www.ebr.edu.pl

#### Subscription

Economics and Business Review (EBR) is published quarterly and is the successor to the Poznań University of Economics Review. The EBR is published by the Poznań University of Economics and Business Press.

Economics and Business Review is indexed and distributed in Scopus, Claritave Analytics, DOAJ, ERIH plus, ProQuest, EBSCO, CEJSH, BazEcon, Index Copernicus and De Gruyter Open (Sciendo).